Author's response to reviews

Title: The influence of socio-cultural interpretations of pregnancy threats on health-seeking behavior among pregnant women in urban Accra, Ghana

Authors:

Phyllis Dako-Gyeke (gyekenay@yahoo.com)
Philip Baba Adongo (adongophilip@yahoo.com)
Moses Aikins (maikins57@yahoo.com)
Richmond Aryeetey (rnokai@yahoo.co.uk)

Version: 3  Date: 2 November 2013

Author's response to reviews:

Authors' contributions

PDG participated in data collection and qualitative data analysis of the study. PBA participated in the design of the study. MA participated in the design, undertook the cost analysis and wrote the initial manuscript of the study. RA helped in the initial proposal development and coordinated the data collection and management and data analysis. All the authors were involved in drafting, revising the manuscript, reading and approving the final manuscript.